Author’s response to reviews

Title: Emergence of Endemic Serogroup C Meningococcal Disease Associated with a High Mortality Rate in Hefei city, China

Authors:

Xu Xi-Hai (2xhai@163.com)
Ye Ying (yeying2009@163.com)
Hu Li-Fen (lifenhu@gmail.com)
Jiang Qin-Qin (jiangqq2010@163.com)
Jin Yu-Hui (yhuijin@sina.com.cn)
Li Jia-Bin (jiajibinli948@163.com)

Version: 3  Date: 27 October 2011

Author’s response to reviews: see over
Cover Letter

Dear Roselle Pangilinan,

Good day.

Thank you for your letter about the questions of our manuscript. According to the requirement, we have made some revisions to the manuscript.

1. We submitted the information about ethics statement in the Methods section of the manuscript. The protocol of the study adhered to the tenets of the Declaration of Helsinki and was approved by the ethics committee of Anhui Medical University.

2. We placed the authors' information in the Authors' Contributions section.

3. We added the Competing interests 'Competing interests' section and provided page numbers throughout our manuscript.

Best regards,

Yours sincerely,

Jia-Bin Li

Department of Infectious Diseases,
the First Affiliated Hospital of Anhui Medical University,
Hefei, Anhui 230022, the People’s Republic of China.